Author's response to reviews

Title: A displaced stress fracture of the femoral neck in an adolescent female distance runner with female athlete triad: A case report

Authors:

Shinichi Okamoto (okamoty@koto.kpu-m.ac.jp)
Yuji Arai (yarai89046@nike.eonet.ne.jp)
Kunio Hara (acl812hara@yahoo.co.jp)
Takashi Tsuzihara (tuzihara@gakusai.or.jp)
Toshikazu Kubo (tkubo@koto.kpu-m.ac.jp)

Version: 3 Date: 4 March 2010

Author's response to reviews:

Dear the editor of SMARTT Journal,

We resubmit a case report named “A displaced stress fracture of the femoral neck in an adolescent female distance runner with female athlete triad: a case report” (MS: 4369502413234610).

We corrected the major and minor revisions which you pointed out.

We are looking forward to hearing from you. Sincerely yours,

Shinichi Okamoto
Dept. of Orthopaedics, Graduate School of Medical Science, Kyoto Prefectural University of Medicine, Kawaramachi-Hirokoji, Kamigyo-ku, Kyoto 602-8566, Japan
Tel: +81-75-251-5549
Fax: +81-75-251-5841
E-mail: okamoty@koto.kpu-m.ac.jp